Author's response to reviews

Title: Measuring the burden of arboviral diseases: the spectrum of morbidity and mortality of four prevalent infections

Authors:

A Desiree LaBeaud (alabeaud@chori.org)
Fatima Bashir (drfatima08@hotmail.com)
Charles H King (chk@case.edu)

Version: 4 Date: 4 January 2011

Author's response to reviews:

I have made the requested proofing changes.